Author's response to reviews

Title: Factors related to functional prognosis in elderly patients after accidental hip fractures. A prospective cohort study.

Authors:

Itziar Vergara (ivergara@osakidetza.net)
Kalliopi Vrotsou (KALLIOPI.VROTSOUKANARI@osakidetza.net)
Miren Oribe (MIREN.ORIVECALZADA@osakidetza.net)
Nerea Gonzalez (NEREA.GONZALEZHERNANDEZ@osakidetza.net)
Susana Garcia (SUSANA.GARCIAGUTIERREZ@osakidetza.net)
JoseMaria Quintana (JOSEMARIA.QUINTANALOPEZ@osakidetza.net)

Version: 2 Date: 22 July 2014

Author's response to reviews:

Dear Sir/Madam,

In a preliminary review I have been asked to make the following changes in the manuscript:

1. to include Authors' Contributions information
2. to include a Title Page.
3. to include an Acknowledgement section
4. to include page and Line Numbering

All these changes have been made in the new manuscript version I am uploading now.

I am very sorry for the inconvenience and thankful for your help.

Best regards,

Itziar Vergara